Author’s response to reviews

Title: Talaromyces marneffei infection in a lung cancer patient: a rare case report

Authors:

Fanhai Lin (1667651741@qq.com)
Ye Qiu (287063367@qq.com)
Wen Zeng (327627266@qq.com)
Yi Liang (448928468@qq.com)
Jianquan Zhang (jqzhang2002@sina.com)

Version: 2 Date: 29 Mar 2019

Author’s response to reviews:

29 Mar 2019

Antoni Torres
BMC Infectious Diseases Editorial Office
BMC Infectious Diseases

Dear Dr. Torres,

Thank you for reviewing our manuscript. We have carefully considered all of the comments and modified the manuscript accordingly. The details of our revisions are explained in our point-by-point response below.

Comment 1: Please combine your declarations statements entitled "Ethics approval" and "Consent to Participate" into one heading entitled "Ethics approval and consent to participate" as per BMC Infectious Diseases submission guidelines.

Answer: We have combined declarations statements entitled "Ethics approval" and "Consent to Participate" into one heading entitled "Ethics approval and consent to participate". 
Comment 2: Please amend your "Availability of Data and Materials" statement to read "Not Applicable".

Answer: We have revised the "Availability of Data and Materials" statement to read "Not Applicable".

Comment 3: Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Answer: We have added the following sentence: "All authors read and approved the final manuscript." (Page 10, line 2).

Comment 4: Please remove any additional files you do not want published alongside your manuscript, including cover/response letters and certificates.

Answer: We have removed the previously uploaded response letters and certificates.

Comment 5: At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Answer: The manuscript has been modified as a single, final, clean version. A figure has been uploaded as a single part.

Sincerely,

Fanhai Lin
Corresponding author:
Jianquan Zhang
E-mail: jqzhang2002@sina.com